Author's response to reviews

Title: Does point of care prothrombin time measurement reduce the transfusion of fresh frozen plasma in patients undergoing major surgery? The POC-OP randomized-controlled trial

Authors:

Natalie Urwyler (natalie.urwyler@insel.ch)
Sven Trelle (strelle@ctu.unibe.ch)
Lorenz Theiler (lorenz.theiler@insel.ch)
Peter Juni (juni@ispm.unibe.ch)
Lukas P Staub (lukas.staub@ctc.usyd.edu.au)
Cedric Luyet (cedric.luyet@insel.ch)
Lorenzo Alberio (lorenzo.alberio@insel.ch)
Kay Stricker (kay.stricker@insel.ch)
Robert Greif (robert.greif@insel.ch)

Version: 4 Date: 10 November 2009

Author's response to reviews:

To whom it may concern

In the revised manuscript an author’s contributions section is included. I clarified the figure legends and I checked for typographical errors.

Yours sincerely

Natalie Urwyler